Author's response to reviews

Title: A Meta-analysis of the effects of Exercise Training on Left Ventricular Remodeling Following Myocardial Infarction: Start early and go longer for greatest exercise benefits on remodelling

Authors:

Mark Haykowsky (Mark.Haykowsky@ualberta.ca)
Jessica Scott (Jessica.Scott@ualberta.ca)
Ben Esch (Ben.Esch@ualberta.ca)
Don Schopflocher (Don.Schopflocher@ualberta.ca)
Johnathon Myers (dri93@aol.com)
Ian Paterson (Ian.Paterson@ualberta.ca)
Darren Warbuton (darren.warburton@ubc.ca)
Lee Jones (Lee.W.Jones@duke.edu)
Alexander M Clark (alex.clark@ualberta.ca)

Version: 4 Date: 22 March 2011

Author's response to reviews:

Dear Dr Julian

Thanks for the request to revise the abstract for the above paper to meet with the journal requirements. The abstract is now 350 words and structured in accordance with your request.

Sincerely,

Alex Clark and Mark Haykowsky, for the study authors